Author's response to reviews

Title: Impact of hiatal hernia on histological pattern of non-erosive reflux disease

Authors:

Anthie Gatopoulou (gatop@otenet.gr)
Konstantinos Mimidis (kmimidis@otenet.gr)
Alexandra Giatromanolaki (kmimidis@otenet.gr)
Vassilios Papadopoulos (vaspapmd@mail.otenet.gr)
Alexandros Polychronidis (kmimidis@otenet.gr)
Nikolaos Lyratopoulos (kmimidis@otenet.gr)
Efthimios Sivridis (kmimidis@otenet.gr)
Georgios Minopoulos (kmimidis@otenet.gr)

Version: 5 Date: 24 December 2004

Author's response to reviews:

Dear Editor,

We have proceeded to the following changes as requested:

1. The language has been scanned and improved by P.J. Kolston, PhD, Managing Editor, www.english-science-editing.com.

2. The formatting has been revised according to the instructions given:
   a. Corresponding author has been added
   b. Affiliations have been added
   c. E-mail addresses have been included where available
   d. Abstract has been revised (Background section)
   e. References have been cited in the text using consecutive > numbers in square brackets
   f. Authors' contributions have been noted according to the format requested
   g. Declaration of competing interests has been added
   h. Table has been reformed according to your instructions
   i. Image files have been resubmitted properly cropped without any legends on them

We apologise for any inconvenience that we may have caused to you.

Yours Sincerely,

Konstantinos Mimidis, MD, PhD
Vassilios Papadopoulos, MD, PhD